Author's response to reviews

Title: Assessment of the feasibility of exon 45-55 multiexon skipping for Duchenne Muscular Dystrophy

Authors:

Laura van Vliet (l.van_vliet@lumc.nl)
Christa L de Winter (c.l.de_winter@lumc.nl)
Judith CT van Deutekom (j.vandeutekom@prosensa.nl)
Gert-Jan B van Ommen (gjvo@lumc.nl)
Annemieke Aartsma-Rus (a.m.rus@lumc.nl)

Version: 5 Date: 19 November 2008

Author's response to reviews:

We have made the following changes to the manuscript:

1) We have inserted table 1 after the references as requested
2) We have changed Material and Methods into Methods as requested
3) We have paragraphed rather than listed the abbreviations as requested
4) We have cropped the figures to minimize the white border as requested
5) We have updated the percentages of applicability of single and multiexon skipping on page 3 and page 9, which resulted from re-analyzing our database and added (Aartsma-Rus et al, accepted manuscript) to refer to the upcoming paper that describes this analysis
6) We have changed Heemskerk et al submitted manuscript to accepted manuscript, as this manuscript has now been accepted
7) We have changed oligomer to oligomers (typo) on page 5, 6th sentence from the bottom